Author’s response to reviews

Title: Novel BRCA2 pathogenic variant c.5219T>G p.(Leu1740Ter) in a consanguineous Senegalese family with hereditary breast cancer

Authors:

Jean Diop (diopjeanpascaldemba@gmail.com)
Rokhaya Ndiaye Diallo (dabafr@yahoo.fr)
Violaine Bourdon-Huguenin (BOURDONV@ipc.unicancer.fr)
Ahmadou Dem (adehdem@gmail.com)
Doudou Diouf (doudougaradiouf@gmail.com)
Mamadou Dieng (moustaphamamadou@gmail.com)
Seydi Abdoul (banourrou@gmail.com)
Yacouba Dia (diayacoub87@yahoo.com)
Sidy Ka (sidy.ca.ucad.edu.sn)
Babacar Mbengue (bmbengue@yahoo.fr)
Alassane Thiam (thiam@pasteur.sn)
Oumar Faye (doffaye@yahoo.fr)
Papa Diop (prpadiop@yahoo.fr)
Hagay Sobol (SOBOLH@ipc.unicancer.fr)
Alioune Dieye (alioune.dieye@ucad.edu.sn)

Version: 3 Date: 17 Apr 2019

Author’s response to reviews:

1. In the Ethics approval and consent to participate section, please clearly state if consent to participate was written or verbal.

We have clarified that it was a written consent to participate
2. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

We added in the appropriate section that all authors have read and approved the manuscript.

3. Please ensure all authors contributions are up to date.

All authors contributions are up to date

4. Thank you for providing your response to reviewers and cover letter. However, at this stage it is not required and so we kindly ask that you remove it from your manuscript.

We have removed the cover letter and the response to reviewers from the manuscript